Author's response to reviews

Title: In vitro study on the schedule-dependency of the interaction between pemetrexed, gemcitabine and irradiation in non-small cell lung cancer and head and neck cancer cells.

Authors:

An Wouters (an.wouters@ua.ac.be)
Bea Pauwels (bea.pauwels@ua.ac.be)
Filip Lardon (filip.lardon@ua.ac.be)
Greet GO Pattyn (greet.pattyn@ua.ac.be)
Hilde AJ Lambrechts (hilde.lambrechts@ua.ac.be)
Marc Baay (marc.baay@ua.ac.be)
Paul Meijnders (paul.meijnders@zna.be)
Jan B Vermorken (jan.vermorken@ua.ac.be)

Version: 5 Date: 18 August 2010

Author's response to reviews: see over
Dear Prof. Norton,

We are very pleased that our revised manuscript has been accepted for publication in BMC Cancer and we want to thank you very much for this. Please find enclosed the final version of our manuscript, in which we addressed the last amendments to the format of our paper, in detail:

- Tables should be included and visible in the final published manuscript. Therefore, tables were retyped and included in a table section following the references. We hope the formatting is now conforming the journal guidelines. If not, do not hesitate to contact us.

- All highlighting and tracking are removed from the manuscript.

- The reference list now contains up to 30 authors’ names for each citation.

- The manuscript was checked for typographical errors.

Yours sincerely,
An Wouters